Author’s response to reviews

Title: Relationship between oral environment and frailty among older adults dwelling in a rural Japanese community: A cross-sectional observational study

Authors:

Yoko Hasegawa (cem17150@dent.niigata-u.ac.jp)
Ayumi Sakuramoto (ayu.cherry.ayu@gmail.com)
Hideyuki Sugita (hi-sugita@hyo-med.ac.jp)
Kana Hasegawa (ka-hasegawa@hyo-med.ac.jp)
Nobuhide Horii (no-horii@hyo-med.ac.jp)
Takashi Sawada (ahfb4801@bca.bai.ne.jp)
Ken Shinmura (ke-shimmura@hyo-med.ac.jp)
Hiromitsu Kishimoto (kishihiro@hyo-med.ac.jp)

Version: 4  Date: 10 Jan 2019

Author’s response to reviews:

January 10, 2019
Dr. Shenuka Singh
Editor
BMC Oral Health

Dear Editor,

We are resubmitting our revised manuscript, titled “Relationship between oral environment and frailty among older adults dwelling in a rural Japanese community: A cross-sectional observational study” (OHEA-D-18-00353R3) by Yoko Hasegawa, Ayumi Sakuramoto, Hideyuki Sugita, Kana Hasegawa, Nobuhide Horii, Takashi Sawada, Ken Shinmura, and Hiromitsu Kishimoto for publication as an original report in BMC Oral Health.
We are grateful to the reviewers for their time. Following reviewer comments, uploaded plain text and tables.

Respectfully yours,

Yoko Hasegawa
Division of Comprehensive Prosthodontics
Niigata University Graduate School of Medical and Dental Sciences
5274, Gakkocho-dori 2-bancho, Chuo-ku, Niigata, 951-8514 Japan
Tel: +81-25-227-2891
Fax:+81-25-229-3454
mail: cem17150@dent.niigata-u.ac.jp